Author's response to reviews

Title: Canadian physiotherapists’ views on certification, specialization, extended role practice, and entry-level training in rheumatology

Authors:

Linda C Li (lli@arthritisresearch.ca)
Marie D Westby (Marie.Westby@vch.ca)
Evelyn Sutton (evelyn.sutton@cdha.nshealth.ca)
Marlene Thompson (marlene.thompson@rogers.com)
Eric C Sayre (esayre@arthritisresearch.ca)
Lynn Casimiro (casimiro@uottawa.ca)

Version: 5 Date: 26 May 2009

Author's response to reviews:

May 25, 2009

Dear BioMed Central Editorial Production Team,

RE: Canadian physiotherapists’ views on certification, specialization, extended role practice, and entry-level training in rheumatology

Enclosed please find the revision of the abovementioned paper. The following changes have been made:

1. All affiliations are now written in full.
2. The old references #12 and 41 have been removed. The reference list has been updated
3. Table 1 has been reformatted using the Tables tool.
4. All table shading and colour fonts have been removed.
5. All table citations are also mentioned within the main text.
6. A functioning web link has been included for the old reference #44 (now #30).

Sincerely yours,

Linda Li